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D3R Status
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Date:
Thursday, May 9, 2013 - 23:15

Instrument List:

e Green
Discussion:

D3R oprated fairly well today. Scans were stopped briefly to investigate fan failure on one of
the thermoelectric coolers on the IF box. After inspection, a small rock was found inside fan
cage which may have jammed one of the fans. Spare fans were ordered (shipped to Mike
Watson to be delivered on 5/13) just in case the current ones may need to be replaced in the
near future.

BB scans performed at: 14:29, 17:51 and 18:11 UTC.
MVega and HChen



